Author's response to reviews

Title: Ebsteins Anomaly: Anatomo-echocardiographic correlation

Authors:

Luis Munoz-Castellanos Professor (lmunoz@hotmail.com)
Nilda Espinola-Zavaleta M.D, PhD (niesza2001@hotmail.com)
Magdalena Kuri-Nivon M.D. (magerep@hotmail.com)
Candace Keirns M.D. (mieshe@comcast.net)

Version: 2 Date: 11 November 2007

Author's response to reviews: see over
Mexico city, 12 November, 2007

Eugenio Picano
Editor in Chief of the “Cardiovascular Ultrasound”

Dear Dr Picano

We are sending the revised manuscript entitled “Ebstein’s Anomaly: Anatomoechocardiographic correlation” according to the reviewere’s suggestions:

1. We aligned the title to the left
2. We removed the authors’ titles from the manuscript and submission system
3. Affiliations for each author below the author list were provided
4. The running title and keyword section were removed
5. The all email authors was provided
6. The underlining in the final manuscript was changed to bold
7. The references were cited in the text using consecutive numbers in square brackets
8. Material and methods was renamed to Methods
9. The tables are visible in the manuscript and we refer to these as table 1 and table 2
10. We minimized white space around the all images

Sincerely yours,
Nilda Espinola-Zavaleta, M.D., PhD
Echocardiography in Out-Patients Clinic, INC “Ignacio Chavez”